Author's response to reviews

Title: Molecular epidemiology of pneumococci obtained from Gambian children aged 2-29 months with invasive pneumococcal disease during a trial of a 9-valent pneumococcal conjugate vaccine

Authors:

Martin Antonio (mantonio@mrc.gm)
Hannah Dada-Adegbola (dadaadegbola@yahoo.com)
Ekow Biney (ekowbee746@yahoo.com)
Tim Awine (tawine@mrc.gm)
John O'Callaghan (ocallaghan.john@gmail.com)
Valentin Pfluger (V.Pflueger@stud.unibas.ch)
Godwin Enwere (genwere@mrc.gm)
Brown Okoko (bokoko@mrc.gm)
Claire Oluwalana (colulana@mrc.gm)
Adeola Vaughan (ade_vaughan@hotmail.com)
Syed M.A Zaman (azaman228@yahoo.com)
Gerd Pluschke (gerd.pluschke@unibas.ch)
Brian M Greenwood (Brian.Greenwood@lshtm.ac.uk)
Felicity Cutts (Felicity.Cutts@lshtm.ac.uk)
Richard A Adegbola (radegbola@mrc.gm)

Version: 8 Date: 8 June 2008

Author's response to reviews:

8th June 2008

The BioMed Central Editorial Production Team

Dear Editorial Production Team

Re: MS: 4915439411580049
Thank you for your e-mail of 5th June 2008 concerning the above manuscript.

In response to your formatting changes, please note that

1. The references are submitted as RTF
2. Title of 'Figures' is changed to 'Figure legends'.
3. Table shading have been removed
4. Figures are saved as 178x250 in PPT

I attached the revised manuscript, the tables and figures. Thank you for your kind consideration.
Yours sincerely,
Dr Martin Antonio
(On behalf on all authors)